Author's response to reviews

Title: Using Cognitive Theory to Facilitate Medical Education

Authors:

Yu Qi Qiao (medqyq@gmail.com)
Jun Shen (junshen1979@qq.com)
Xiao Liang (xiao_liang001@qq.com)
Song Ding (songding1981@qq.com)
Fang Yuan Chen (fangyuan_chen@qq.com)
Li Shao (lishaoo@qq.com)
Qing Zheng (qingzheng101@qq.com)
Zhi Hua Ran (zhihuaran@vip.126.com)

Version: 5  
Date: 19 March 2014

Author's response to reviews:

Dear Editors,

Small changes have made. I hope my corresponding email to be zhihuaran@vip.163.com when print. You can still contact me with zhihuaran@vip.126.com.

Kind Regards,

Zhi Hua Ran